Author's response to reviews

Title: Estimating prevalence of overweight and obesity at the neighborhood level: the value of maternal height and weight data available on birth certificate records.

Authors:

David A Webb (webbd@email.chop.edu)
Jessica M Robbins (Jessica.Robbins@Philagov.org)
Joan R Bloch (culhanej@email.chop.edu)
Jennifer F Culhane (culhanej@email.chop.edu)

Version: 7 Date: 27 April 2010

Author's response to reviews:

We wish to have one small change made in the manuscript. On page 9 the last sentence should read and was replaced by:

"Since the maternal height and weight data elements are new to the revised Standard Certificate the findings here may be timely, and particularly relevant to city/county public health professional who have a growing need for obesity-related indicators at the small area or neighborhood levels."

All other sections of the manuscript, Tables and Figure 1 have been reviewed and we have found no problems